Author's response to reviews

Title: Genetic Integrity of the human Y chromosome exposed to groundwater arsenic

Authors:

Safdar Ali (alisafdar@nii.ac.in)
Sher Ali (alisher@nii.res.in)

Version: 2 Date: 8 February 2010

Author's response to reviews:

The Editor
BMC Medical Genomics

Dear editor,

Enclosed alongwith is a research manuscript entitled 'Genetic integrity of the human Y chromosome exposed to groundwater arsenic' for publication in BMC Medical Genomics, should you find it appropriate. The article was originally submitted to Genome Biology and on the recommendation of its editorial board we are resubmitting it to your esteemed journal.

Both the authors Safdar Ali and Sher Ali have read the manuscript and there are no conflicts of interest for the same.

We would also like to confirm that should the manuscript be accepted for publication we are willing to pay the publication charges as per the journal guidelines.

Thanking you

Dr. Sher Ali
Head, Molecular Genetics Lab
NII, Aruna asaf Ali Marg
New Delhi-110067